Contents lists available at ScienceDirect

The Journal of Foot & Ankle Surgery

journal homepage: www.jfas.org

Pain Reduction With AbobotulinumtoxinA for the Treatment of Hallux Valgus in Adult Participants: Results of a Randomized and Placebo-Controlled Phase 2 Trial

David G. Armstrong, DPM, MD, PhD¹, Lawrence A. DiDomenico, DPM, FACFAS², Babak Baravarian, DPM³, Selene G. Parekh, MD, MBA⁴, Magali Volteau, MSc⁵, Robert Silva, PhD⁶

¹ Keck School of Medicine, University of Southern California, Los Angeles, CA

² NOMS Ankle and Foot Care Centers, Youngstown, OH

³ University Foot and Ankle Institute, Los Angeles, CA

⁴ Duke University Medical Center, Durham, NC

⁵ Ipsen, Les Ulis, Paris, France

⁶ Ipsen, Cambridge, MA

ARTICLE INFO

Level of Clinical Evidence: 1

Keywords: botulinum toxin foot pain nonsurgical treatment number needed to treat pain rating scale safety

ABSTRACT

AbobotulinumtoxinA (aboBoNT-A, Dysport® [Ipsen, Paris, France]) inhibits acetylcholine release at the neuromuscular junction and may modulate pain signaling in hallux valgus (HV). This randomized study (NCT03569098) included a double-blind phase (aboBoNT-A 300U, 500U or placebo injections into forefoot muscles) and an open-label aboBoNT-A treatment period in participants with an HV diagnosis and no HV surgery. The primary endpoint was change from baseline in numeric pain rating scale (NPRS) score at week 8. Secondary endpoints included change in NPRS (other time points) and proportion of participants with \geq 20% reduction from baseline NPRS (responders). Post-hoc analyses assessed number of days in a 7-day evaluation period that participants spent in a lower pain state than at baseline. Participants received abo-BoNT-A 300U (n = 63), 500U (n = 60) or placebo (n = 63). Superiority to placebo was not observed with either aboBoNT-A dose at week 8, thus the primary endpoint was unmet. At week 12, a trend toward efficacy was observed with aboBoNT-A 500U versus placebo and the proportion of participants with \geq 20% reduction from baseline NPRS was greater with aboBoNT-A 500U versus placebo (p = .006). Participants in the aboBoNT-A 500U group spent more days with lower NPRS than their lowest baseline score, and with NPRS ≥ 2 points lower than their mean baseline NPRS at weeks 8 and 12 versus placebo (all p < .05; posthoc). AboBoNT-A was well tolerated. Although the primary endpoint was unmet, other endpoints showed a nominal advantage for aboBoNT versus placebo for treatment of HV-related pain, particularly at week 12. Further clinical evaluation is needed to establish whether botulinum toxins represent a viable non-operative treatment option for HV-associated pain.

Plain language summary: Hallux valgus is the medical name for a bunion, a foot deformity that can worsen over time. Patients with bunions experience pain and walking can become difficult, which can affect their quality of life. Foot support aids (e.g., braces, splints and inserts) are available, but surgery is the standard treatment.

This study looked at how injections of a specific type of botulinum toxin, called abobotulinumtoxinA or "aboBoNT-A", into the foot may help to reduce pain in patients with bunions. The study included 186 patients aged 18 to 75 years who had not had surgery on their bunion. The researchers looked at how well the injections worked using scales that measure the pain levels the patient experienced.

The main outcome was whether patients who had aboBoNT-A injections had less pain after 8 weeks than they did before treatment. The study included patients who were injected with saltwater (no treatment) to check that any treatment effect was real. Researchers also looked at the results after 12 weeks, as well as how many patients had

Financial Disclosure: The study was sponsored by Ipsen, who were involved in the study design, analysis and interpretation of results.

Conflict of Interest: DGA and BB are investigators for the current study and report consultancy (advisory board) for Ipsen. LAD is an investigator for the current study. SGP has no disclosures to declare. MV is an Ipsen employee. RS was an Ipsen employee during the study and manuscript preparation.

Address correspondence to: Robert Silva, PhD, 23A Sycamore St. Millbury, MA 01527, USA.

E-mail address: Robsilva2188@gmail.com (R. Silva).

1067-2516/\$ - see front matter © 2022 The Author(s). This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/) https://doi.org/10.1053/j.jfas.2022.07.002







less pain after treatment than before and how many days in a given week patients experienced less pain after treatment than they did before.

There was no reduction in pain levels with aboBoNT-A injections after 8 weeks compared with no treatment. However, the other study outcomes suggested that aboBoNT-A resulted in a small benefit compared with no treatment, especially after 12 weeks. Further medical research is needed to establish whether botulinum toxins represent an alternative treatment to surgery for the pain associated with bunions.

© 2022 The Author(s). This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

Globally, approximately 23% of adults below 65 years of age are afflicted with hallux valgus (HV) (1), a progressive deformity characterized by lateral deviation of the great toe (the hallux) and medial deviation of the first metatarsal head, which causes a prominence commonly known as a "bunion" (2,3). The etiology of HV is thought to involve an imbalance between the abductor (decreased activity) and adductor hallucis muscles, whereby the latter gains a mechanical advantage and the hallux is pulled laterally causing the HV forefoot morphology (4,5). Subsequent structural changes in the joint can lead to significant and chronic pain, often related to extrinsic factors (such as pressure of a shoe over the medial eminence), severe mobility limitations (6), emergence of ancillary pain conditions (e.g., transfer metatarsalgia (6) and sensory nerve dysfunction) (7), and functional impairments in gait and balance (8,9), all of which negatively impact quality of life (10-13).

Together with morphological changes in the foot, forefoot pain and disability are key indicators of HV severity and treatment outcomes (14,15). Early medical management of HV involves orthotic interventions (16), such as braces, splints or inserts. However, with limited effect in clinical trials in correcting the biomechanics of the foot, reducing HV-related pain or preventing HV progression, orthoses may be no more effective than no treatment at all (16-18). Currently, the standard of care is operative correction, which generally includes rebalancing soft tissue, reducing angular deviation, and realigning the articular surface of the metatarsal (6,17). However, postoperative pain and swelling, a recovery period of up to 12 weeks (6) and a risk of recurrence (5,19) complicate risk/benefit decisions regarding surgery for participants with HV. There is a dearth of effective non-operative treatments for HV for those who cannot, or who choose not to, undergo operative procedures.

Botulinum toxins (BoNTs) are neuromuscular blocking agents that inhibit release of presynaptic acetylcholine at the neuromuscular junction, reducing localized muscle tone in the injected muscle (20). Pain signaling in the periphery, the dorsal root ganglia and spinal cord is also impaired following BoNT injection, leading to peripheral and central pain desensitization (20-22). BoNTs have reliably shown efficacy in the treatment of pain conditions such as migraine (23), trigeminal neuralgia (22), hypertrophic scar pain (24) and plantar fasciitis (25). Prior studies in participants with HV have demonstrated robust, clinically meaningful changes in foot pain and functional mobility following direct injection of BoNT type A (BoNT-A) into specific forefoot muscles governing the hallux and related metatarsal structures (26-28). In addition to reducing pain, there is some evidence that BoNT injections can reduce HV-associated angular deviation owing to the temporary paralysis of muscle and subsequent elimination of one of the deforming forces (26-28).

AbobotulinumtoxinA (aboBoNT-A, Dysport[®] [Ipsen, Paris, France]) is approved in several countries for the treatment of spasticity, cervical dystonia and facial esthetics (29-31). Based on its mechanism of action, it was hypothesized that aboBoNT-A would reduce pain in participants with HV. The primary aim of the present randomized controlled study was therefore to assess pain reduction in adults with HV who had not undergone surgery, following injections of aboBoNT-A 300U and 500U compared with placebo, using the numeric pain rating scale (NPRS) (32,33). Secondary objectives were to assess the effect of aboBoNT-A on functional impairment, activity limitation, angular displacement, quality of life and participants' global impressions of severity and improvement, and to evaluate the safety and tolerability of aboBoNT-A in patients with HV who had not undergone surgery.

Participants/Materials and Methods

Study Design

This was a randomized, double-blind, placebo-controlled, parallel-group, multicenter, phase 2 study (NCT03569098 [ClinicalTrials.gov]; Fig. 1). The total duration was 39 weeks, and comprised a double-blind period of at least 12 weeks (cycle 1) and an open-label period lasting up to 24 weeks (cycles 2 and 3). The study was approved by appropriate health authorities and by independent ethics committees/institutional review boards, and was conducted in accordance with the Declaration of Helsinki, the International Conference on Harmonization Consolidated Guideline on Good Clinical Practice and all local regulatory guidelines. Written informed consent was provided by all participants prior to enrollment.



Fig. 1. Study design.

Abbreviations: AboBoNT-A, abobotulinumtoxinA; BL, baseline; FU, follow-up; N, number of participants; U, units.

Randomization and Blinding

Participants, investigators and study and sponsor personnel were blinded to treatment assignment throughout the study. All treatments were similar in size, color, smell, taste and appearance, allowing the blinded conditions of the study to be maintained. The placebo treatment (saline) was provided in glass vials and was indistinguishable from the active product.

Randomization numbers were produced in blocks on a balanced ratio (1 placebo:1 aboBoNT-A 300U:1 aboBoNT-A 500U) and stratified by unilateral and bilateral HV. After eligibility was confirmed at baseline, participants were assigned a randomization number and allocated to the associated treatment arm. Randomization and assignment of treatment number were managed by an independent person using an interactive response technology system.

Participants

Participants were recruited at each investigator's institution, as listed in Supplementary Appendix 1. Supplementary Appendix 2 lists full inclusion and exclusion criteria. In order to ensure that variability was minimized and potential confounding factors eliminated, the entry criteria restricted the inclusion of participants with pre-existing medical conditions and those reliant on treatments not permitted in the study. Key inclusion criteria were: age 18 to 75 years; unilateral or bilateral HV diagnosis; HV angle <30 degrees; first intermetatarsal (IM) angle <18 degrees associated with the hallux of the study foot; reducible deformity (investigator judgment); and foot pain refractory to shoe modifications, non-steroidal anti-inflammatory medications and activity modification. Participants must have recorded scores of ≥ 4 on the NPRS and >27 on the modified foot function index (mFFI) pain and disability subscales in the study foot. Key exclusion criteria included: an inability to walk unassisted; any other podiatric/orthopedic condition interfering with pain and/or function evaluation; history of ankle or foot surgery in the study foot; diabetes, peripheral neuropathy, inflammatory arthritis, osteoarthritis, disease causing ligamentous laxity; or a body mass index >40 kg/m² or <18.5 kg/m². Use of orthotic inserts/devices on the study foot (except over-the-counter orthoses used for 30 days prior to screening) and BoNT treatment <4 months prior to screening for any condition (except esthetic facial applications) were not permitted. Participants with severe HV (HV angle \geq 30 degrees or IM angle \geq 18 degrees) were excluded because they are potential candidates for surgery; also, the upper limit on the HV angle was anticipated to reduce the need for "rescue" surgery. Participants with fixed deformity were excluded because it was postulated that BoNT treatment may improve structural disease, but only if the HV was semi-reducible or reducible.

Treatment

In the double-blind period (cycle 1), participants were randomized 1:1:1 to receive aboBoNT-A 300U, aboBoNT-A 500U or placebo (Fig. 1). These doses were based on the demonstrated safety and efficacy of aboBoNT-A in reducing hypertonia in muscles of similar size and volume to those planned for evaluation in the current study (34,35). Twelve weeks after treatment in cycle 1, participants from each group who met retreatment criteria entered the open-label period and received aboBoNT-A 300U (cycle 2), regardless of treatment received in the double-blind period. Participants who met retreatment criteria again at least 12 weeks later received either aboBoNT-A 300U or aboBoNT-A 500U (investigator judgment) in cycle 3. Retreatment criteria were: participant consent; investigator's clinical judgment; clinically significant foot pain (NPRS ≥3) in the preceding 24 hours; and no unacceptable risk experienced by the participant (investigator judgment). The decision to increase the dose to 500U at the beginning of cycle 3 was based on an evaluation of safety and tolerability (treatment-related adverse events [AEs] and any significant changes in the study foot), severity of pain (NPRS score) and disability (mFFI disability subscale score). From week 12 of each treatment cycle, participants not meeting these criteria were clinically re-evaluated every 4 weeks to assess eligibility for retreatment

Each participant received up to 3 administrations (i.e., cycles) of study treatment (one double-blind and up to 2 open-label), each with a minimum 12-week follow-up period, to assess the effect of repeated treatment cycles. For each cycle, study treatments were administered by the investigator with assistance from sub-investigators (if relevant) at each clinical site, and consisted of a single set of 4 intramuscular injections to the study foot of aboBoNT-A 75U (total dose 300U), aboBoNT-A 125U (total dose 500U) or placebo. The total dose of aboBoNT-A was divided equally among the oblique and transverse heads of the adductor hallucis muscle, the flexor hallucis brevis muscle and the extensor hallucis brevis muscle. The appropriate target muscles were identified using a peripheral electrical stimulator, with or without complementary techniques. A Teflon-coated, 27- to 30gauge, open-lumen needle was used to stimulate the target muscle once per second (repetitive square wave pulses, 0.25 msec in duration), and injection was performed when either a continuous or stretch of muscle was located. For participants with bilateral HV, only the foot with the most severe pain, based on NPRS and clinical evaluation at baseline, was treated and evaluated. The same investigator at each site administered interventions and assessed for retreatment eligibility (DGA, LAD, BB), with assistance from sub-investigators when relevant.

Assessments

For all 3 cycles, study visits were conducted at baseline and weeks 1, 4, 8 and 12, with additional visits conducted every 4 weeks thereafter in cycles 1 and 2 to assess eligibility for retreatment, as needed. Participants, who were blinded to treatment allocation, used electronic diaries to self-record daily NPRS and mFFI scores for 7 consecutive days prior to baseline and at each post-baseline study visit. The NPRS is an 11-point scale (0-10, no pain to worst possible pain) (36). The mFFI consists of 21 items (modified from 23 items (37,38) to exclude 2 relating to use of assisted devices) grouped into 3 11-point subscales: pain (0-10, no pain to worst possible pain), disability (0-10, no difficulty to so difficult unable) and activity limitation (0-10, none of the time to all of the time). Patient Global Impression of Improvement (PGI-I) and Patient Global Impression of Severity (PGI-S) scores for disability and pain were reported by the participants on site at weeks 4, 8 and 12. The PGI-S (4-point Likert scale [0: no pain; 3: severe pain]) and PGI-I (7-point Likert scale [-3: very much worse; 3: very much improved]) scales, respectively evaluate severity of a given symptom and the degree to which it has improved following treatment (39-41). Functional outcomes and perceived well-being in mental, social and physical aspects of life were measured using the 36-item Short-Form Health Survey (SF-36) (42-44), completed by participants at baseline and weeks 8 and 12 in each treatment cycle. Radiographic measurements were based on guidelines set forth by the American Orthopedic Foot and Ankle Society ad hoc Committee on Angular Measurements (45). HV and IM angles were measured directly from weightbearing anteroposterior radiographs, with the X-ray beam angled at 15 degrees toward the heel centered on the second tarsometatarsal joint. Images were to be taken by the same radiology technician at each site, who was not involved in the administration of aboBoNT-A and was blinded to treatment allocation. Angle measurements were performed by a blinded central reader.

Any AEs were reported. AEs that suggested a possible remote spread of the effect of the toxin (Supplementary Appendix 3) or hypersensitivity were of special interest.

Objectives and Endpoints

The primary objective was to assess pain reduction with aboBoNT-A compared with placebo using the NPRS. Secondary objectives were to assess: functional improvement (mFFI disability subscale); foot pain reduction (mFFI pain subscale); activity limitation (mFFI activity limitation subscale); quality of life (SF-36); hallux angular displacement (radiographs); PGI-I and PGI-S for foot pain and disability; and safety and tolerability of aboBoNT-A.

For NPRS and mFFI, mean scores were calculated for the 7 days prior to each time point. Participants must have completed at least 4 of the 7 daily scores for diaries to be considered valid. The primary endpoint was change from baseline in foot pain measured by mean NPRS at week 8 in cycle 1. This time point was chosen because it reflects good-quality care and follow-up in general clinical practice. Secondary endpoints included the change from baseline in mean NPRS at all other time points, the proportion of responders (participants who reported \geq 20% reduction from baseline in mean NPRS), change from baseline in mean mFFI, PGI and SF-36 scores, change from baseline in HV and IM angle, and time to retreatment. To evaluate the number of days in a 7-day evaluation period that participants spent in a "reduced pain state" following treatment, 2 post-hoc analyses were performed. The first assessed the number of days in a 7-day evaluation period that participants reported an NPRS that was lower than their lowest (lowest degree of pain) daily baseline NPRS; the second assessed the number of days in a 7-day evaluation period that participants reported an NPRS tat least 2 units lower than their individual mean NPRS at baseline.

Statistical Analyses

All statistical analyses were conducted on the intent-to-treat population (all randomized participants). To demonstrate superiority of each aboBoNT-A dose compared with placebo, a sample size of 165 randomized participants was estimated assuming a 1.5 point difference in mean change from baseline in NPRS at week 8 between aboBoNT-A and placebo (based on previously reported minimum clinically important difference [MCID] values, as measured by the NPRS in post-bunionectomy pain trials) (32,33,46), a treatment group ratio of 1:1:1, a common standard deviation of 2.5, a power of 80% and a one-sided type I error rate of 2.5%. There was no control of type I error rate for secondary endpoints.

For the primary endpoint, a mixed model for repeated measures (MMRM) was used. Visit (weeks 4, 8 and 12), treatment-group-by-visit interaction, the stratification parameter (unilateral and bilateral HV) and the baseline value were included in the model. For participants without available post-baseline scores, week 8 data were imputed by the mean of the placebo group at week 8 for participants who discontinued because of lack of efficacy, or by the mean of the participants treatment group at week 8 for those who discontinued for other reasons or completed the study but had no valid post-baseline diaries.

An interim analysis was performed after the first 110 randomized participants had been followed for at least 12 weeks to assess both futility and potential for early stopping due to efficacy for each aboBoNT-A group compared with placebo. The overall type I error for each comparison was controlled at the one-sided 0.025 level using O'Brien Fleming spending per Lan-DeMets spending function specification (47). Treatment comparison with an O'Brien Fleming spending corresponds to a nominal one-sided alpha of 0.0062 at the interim analysis. Significance levels were adjusted using the Hochberg procedure to reduce the risk of false positive results due to multiple comparison, under the following decision rules: (1) if the larger of the 2 p values is less than .0062, stop and declare evidence of effect for each arm; and (2) if the larger of the 2 p values is greater than .0062 and the smaller of the 2 p values is less than .0031, then conclude evidence of effect for the arm with the smallest p value. The nonbinding futility boundary was set to declare futility in the interim analysis if the one-sided p value for a comparison was greater than .30. If futility was concluded for the primary endpoint, the secondary endpoint of mFFI pain subscale score (mean change from baseline in the daily mFFI pain subscale score at week 8) was analyzed using the MMRM model. If a one-sided p value was equal to or lower than .05, evidence of a trend was to be declared for the corresponding arm; otherwise, no evidence of a trend for the corresponding arm was concluded. The interim analysis was conducted and reviewed by an independent data monitoring committee; the statistician and programmer were unblinded at the individual participant level for the interim analysis, remained independent from the study, and did not otherwise participate in any study procedures.

The primary analysis was subsequently conducted after all participants had completed week 12 of the double-blind period, but participant entry to the open-label phase was not contingent on the results. The null hypothesis of "no difference between treatment with aboBoNT-A 300U and 500U versus placebo with respect to change from base-line in mean daily NPRS score at week 8" was tested against the alternative hypothesis of "there is a difference between aboBoNT-A 300U or 500U versus placebo." A Hochberg procedure was applied to control the global type I error at the one-sided 2.5% significance level, using the following decision rules: compare larger p value to .0231; if this is less than .0231, then stop and declare evidence of effect for each arm; if this is less than .01155, then conclude evidence of effect for the treatment arm with the smallest p value.

The proportion of "responder" participants with a clinically significant reduction in pain (defined as $\geq 20\%$ reduction in NPRS from baseline) was calculated for the double-blind phase and analyzed with a logistic regression model, with the treatment group, stratification parameter and baseline value as covariates. For analysis on mFFI, HV and IM angle, an MMRM model was used. An mFFI was considered valid if at least 4 of the diary days had at least 50% of items in each subscale completed. No imputation of missing data was done. For PGI and SF-36 data, an analysis of covariance (ANCOVA) model was used and included treatment group, stratification parameter and the baseline value as covariates. Survival curves are presented for time to retreatment data, and a log-rank test performed to compare placebo with the treated groups. An ANCOVA model was used for analysis of post-hoc endpoints. Descriptive statistics are presented for open-label efficacy data and safety data.

Results

Participant Disposition and Baseline Characteristics

This study was conducted at 31 clinical sites in the United States (Supplementary Appendix 1) from June 2018 to May 2020; 27 sites recruited at least one participant. Overall, 531 participants were screened and 345 were screening failures, of whom 335 did not meet the inclusion criteria. Overall, 186 participants were randomized to receive placebo (n = 63), aboBoNT-A 300U (n = 63) or aboBoNT-A 500U (n = 60) in the double-blind phase (cycle 1; 32 randomized participants were from authors' practices [DGA, n = 1; LDD, n = 6; BB, n = 25]) and 180 were treated (n = 61, 63 and 56, respectively) (Fig. 2). In the open-label phase, 146 participants were eligible for retreatment with aboBoNT-A 300U or aboBoNT-A 500U, respectively. Treatment cycles 1, 2 and 3 were completed by 157 (84.4%), 111 (76.0%) and 57 (89.1%) participants, respectively. Emergency unblinding was not required during the study.

Baseline demographics and disease characteristics were similar across the groups (Table 1). Mean (standard deviation [SD]) age was 48.2 (13.1) years, 91.9% of participants were female, and approximately two-thirds of participants had bilateral HV. The mean (SD) HV and IM angles in the study foot were 20.7 (5.2) degrees and 11.9 (2.4) degrees, respectively. Mean (SD) NPRS at baseline was 6.9 (1.6) and mean (SD) time from diagnosis to first injection was 6.4 (8.7) years.



Fig. 2. Participant disposition (screened participants).

If a participant did not meet criteria for additional treatment, they remained in the current cycle. Retreatment was allowed from week 12 onward. In total, 6 participants were lost to follow-up: 2 participants after week 20 of cycle 1; 1 participant after week 4 of cycle 2; 2 participants after week 8 of cycle 2; and 1 participant after week 4 of cycle 3. Abbreviations: AboBoNT-A, abobotulinumtoxinA; ITT, intent-to-treat.

Baseline demographics and disease characteristics

Characteristic	Placebo (n = 63)	AboBoNT-A 300U (n = 63)	AboBoNT-A 500U (n = 60)	All Participants (N = 186)
Age, mean (SD), y	48.3 (13.2)	48.4(14.0) n = 9583	48.0(12.2) n = 8897	48.2 (13.1)
Female, n (%)	55 (87.3)	p = .5365 60 (95.2) p = .1146	p = .8657 56 (93.3) n = .2598	171 (91.9)
HV status, n (%)		<i>p</i>	p 12000	
Unilateral Bilateral n value versus nlacebo	22 (34.9) 41 (65.1)	21 (33.3) 42 (66.7) n = 8510	$ \begin{array}{r} 19 (31.7) \\ 41 (68.3) \\ n = 7020 \end{array} $	62 (33.3) 124 (66.7)
Time since diagnosis, mean (SD), y p value versus placebo*	5.0 (7.1)	p = .0510 6.7 (9.9) p = .2870	7.5(8.8) p = .0890	6.4 (8.7)
HV angle in degrees, mean (SD) p value versus placebo*	20.6 (5.1)	21.3(5.6) p = .4529	20.2(4.9) p = .6275	20.7 (5.2)
IM angle in degrees, mean (SD) p value versus placebo*	11.8 (2.2)	12.2(2.3) p = .3056	11.8(2.7) p = .9586	11.9 (2.4)
NPRS [‡] , mean (SD) p value versus placebo*	6.6 (1.4)	7.2 (1.6) p = .0304	6.9 (1.7) p = .2997	6.9 (1.6)

Abbreviations: AboBoNT-A, abobotulinumtoxinA; HV, hallux valgus; IM, intermetatarsal; ITT, intent-to-treat; NPRS, numeric pain rating scale; SD, standard deviation.

Data are for the ITT population. * Based on pooled *t*-test.

Based on pooled *t*-test.

Based on Chi-Square test.

[‡] Mean of the daily NPRS scores over the 7 consecutive days prior to the baseline visit. A mean is calculated if there are at least 4 days of e-diary completed.

Cycle 1 (Placebo-Controlled Phase)

A decision to continue with the study was concluded based on the outcome of the interim analysis. A table presenting summary statistics of the primary endpoint can be found in the supplementary material (Supplementary Table 1). The primary analysis includes all participants from the intent-to-treat population. All visits performed at weeks 4, 8 and 12 were included in the statistical model (MMRM analysis). For the primary analysis, superiority was not achieved for either aboBoNT-A dose compared with placebo at week 8 in cycle 1, as measured by the change from baseline NPRS (least-squares [LS] mean [95% confidence interval (CI)]: aboBoNT 300U (N = 63), -1.7 [-2.3, -1.1], p = .77; aboBoNT 500U (N = 63), -2.4 [-3.0, -1.8], p = .21; placebo (N = 60), -2.0 [-2.7, -1.4]) (Fig. 3). Given that the primary endpoint was not met, the study was terminated by the sponsor. This decision was not related to any safety or tolerability issues with aboBoNT-A or any other study-related information other than the primary efficacy endpoint interim analysis. The proportion of responders was not significantly different at week 8 with either dose of aboBoNT-A compared with placebo (odds ratio [95% CI]: abo-BoNT-A 300U, 1.08 [0.53, 2.20], p = .84; aboBoNT-A 500U, 1.53 [0.75, 3.14], p = .24) (Fig. 4).

A trend toward efficacy was observed at week 12 in the aboBoNT-A 500U group, with a greater reduction from baseline in mean NPRS compared with placebo (LS mean [95% CI] change from baseline: aboBoNT-A 300U, -1.6 [-2.2, -1.0], p = .59; aboBoNT-A 500U, -2.4 [-1.6, 0.2], p = .06; placebo, -1.7 [-2.4, -1.1] (Fig. 3). A significantly higher proportion of responders compared with placebo was also observed for the aboBoNT-A 500U dose group (53% vs 28%, respectively; odds ratio [95% CI] 2.873 [1.349, 6.118]; p = .0062 (Fig. 4). No significant differences in responder rates were observed for either dose of aboBoNT-A compared with placebo at weeks 4 and 8 (Fig. 4).



Fig. 3. Change from baseline in NPRS in the double-blind phase.

Results were from a mixed model including week 4, 8 and 12 visits. All participants from the ITT population were included, with any missing values for week 8 imputed. Bars represent standard error.

Abbreviations: AboBoNT-A, abobotulinumtoxinA; ITT, intent-to-treat; LS, least-squares; NPRS, numeric pain rating scale.



Fig. 4. Proportion of participants achieving a clinical response in the double-blind phase^a.

Data are for the ITT population. Bars represent standard error. ^aClinical response is defined as the mean reduction from baseline in NPRS of \geq 20% at each time point. Abbreviations: AboBoNT-A, abobotulinumtoxinA; ITT, intent-to-treat; NPRS, numeric pain rating scale.

There were no significant differences observed between either abo-BoNT-A 300U or aboBoNT-A 500U treatment groups compared with placebo at any time points during the double-blind phase in mean change from baseline in mFFI pain, disability or activity subscales or PGI subscales (baseline values are shown in Supplementary Table 2 and the LS mean [95% CI] change values from baseline are shown in Supplementary Table 3). A significant improvement from baseline was observed for the SF-36 Bodily Pain domain with aboBoNT-A 300U compared with placebo at week 12 (LS mean difference [95% CI] 7.52 [-0.87, 15.92], p < .05; Supplementary Table 3). There were no significant differences from baseline between aboBoNT-A treatment groups and placebo in HV or IM angles at any time point (Supplementary Table 3).

Cycles 2 and 3 (Open-Label Phases)

In open-label cycle 2 (aboBoNT-A 300U only), mean NPRS was further nominally reduced from baseline across all groups (n = 152)

regardless of treatment received in the double-blind phase (Fig. 5; Supplementary Table 4). In cycle 3 (aboBoNT-A 300U or 500U), mean NPRS reduced again in both aboBoNT-A dose groups; the greatest mean reduction from baseline was reported with aboBoNT-A 500U – 300U – 500U administered per cycle; however, the number of participants who required a third treatment cycle within the duration of the study was low (n = 3-18; Supplementary Table 4).

Time to Retreatment

The time to retreatment was statistically significantly longer for aboBoNT-A groups compared with placebo (median time [95% CI] to retreatment with placebo was 12.6 [12.29, 12.86] weeks vs 13.1 [12.57, 16.14] weeks for aboBoNT-A 300U, p = .0189 and 13.1 [12.57, 15.43] weeks for aboBoNT-A 500U, p = .0328) (Supplementary Fig. 1).



Fig. 5. Change from baseline in NPRS across 2 treatment cycles according to treatment assignment. Data are for the ITT population. Bars represent standard error. Note that data for cycle 3 are not shown because participant numbers were low (n = 3 to 18).

Abbreviations: AboBoNT-A, abobotulinumtoxinA; ITT, intent-to-treat; NPRS, numeric pain rating scale.



Fig. 6. Least squares mean number of days during a 7-day evaluation period with (A) "lower than lowest" baseline daily NPRS, and (B) \geq 2-point reduction from mean baseline NPRS. *p < .05 versus placebo; **p < .005 versus placebo.

Participants must have completed at least 4 of the 7 daily NRPS scores for diaries to be considered valid.

Abbreviations: AboBoNT-A, abobotulinumtoxinA; LS, least-squares; N, number of participants; n, number of participants with available data; NPRS, numeric pain rating scale; SE, standard error.

Post-Hoc Efficacy Analyses

Post-hoc analyses demonstrated that aboBoNT-A 500U treatment was associated with a trend toward significantly more time spent in a "reduced pain state" compared with placebo treatment in the doubleblind phase. First, participants treated with aboBoNT-A 500U spent more days in a 7-day evaluation period with a lower NPRS score than their lowest score at baseline compared with placebo at week 8 (LS mean [95% CI] 3.89 [3.00, 4.78] vs 2.22 [1.35, 3.10], respectively; *p* = .0028) and week 12 (3.73 [2.89, 4.57] vs 2.04 [1.22, 2.86], respectively; *p* = .0012) (Fig. 6A). Second, participants spent more days in a 7-day evaluation period with a NPRS score \geq 2 points lower than their mean baseline NPRS score with aboBoNT-A 500U compared with placebo at week 8 (LS mean [95% CI] 3.30 [2.40, 4.21] vs 2.15 [1.26, 3.04], respectively; *p* = .0400) and week 12 (2.92 [2.08, 3.77] vs 1.71 [0.88, 2.54], respectively; *p* = .0208) (Fig. 6B).

The clinical significance of these results can be extended by calculating the number needed to treat (NNT), which is a measure of effect size that answers the question "How many participants would one need to treat with an experimental treatment instead of placebo before encountering one additional outcome of interest?" (48,49). At week 8, comparing aboBoNT-A 500U with placebo, the (1) proportion of participants showing a clinical response (20%) and (2) proportion of participants achieving the MCID for HV (\geq 1.5 reduction in NPRS) corresponded to NNT scores of 9.1 and 5.7, respectively. At week 12, the same NNT analyses (clinical response and MCID) resulted in NNT scores of 4.0 and 4.0, respectively.

Safety

In cycle 1, treatment-emergent AEs (TEAEs) were reported in 36.5%, 41.1% and 36.1% of participants in the aboBoNT-A 300U, aboBoNT-A

500U and placebo groups, respectively, the majority being mild-tomoderate in severity (Table 2). The most common TEAEs (in \geq 4% of participants) were injection-site pain, pain in extremity, hyperkeratosis, muscle spasms and nasopharyngitis. No unexpected or new safety signals were observed with aboBoNT-A. TEAEs considered possibly related to treatment by the investigator were observed in 3 (4.8%; aboBoNT-A 300U), 11 (19.6%; aboBoNT-A 500U) and 5 (8.2%; placebo) participants (Table 2). One participant randomized to the placebo group had a serious AE of aphasia during treatment in cycle 3 with aboBoNT-A 500U. The event was considered severe in intensity and unrelated to treatment, and the participant recovered. One AE of special interest (hypersensitivity) was reported in one participant (1.6%) who received placebo.

Discussion

To our knowledge, this is the first large, double-blind, placebo-controlled, clinical study to evaluate the efficacy and safety of intramuscular foot injections of aboBoNT-A, a potential and alternative nonoperative treatment for pain associated with HV. The study, which was conducted in participants who had not undergone HV surgery, showed no significant difference from baseline in mean NPRS following treatment with aboBoNT-A 300U or 500U compared with placebo at week 8 in cycle 1 (primary endpoint). However, results of secondary and posthoc analyses showed that clinically relevant pain reduction was achieved with aboBoNT-A 500U treatment, as suggested by the findings from the trial showing a greater proportion of participants demonstrating a clinically significant reduction in forefoot pain; as measured by multiple pain assessment endpoints, including responder analysis, evaluation of effectiveness under long-term treatment conditions, as well as an alternative pain endpoint measuring the number of days that patients achieved profound pain reduction (severity lower than clinical

Table 2

Summary of adverse events

Adverse Event	Placebo (n = 61)	AboBoNT-A 300U (n = 63)	AboBoNT-A 500U (n = 56)
TEAEs, n (%)			
Any	22 (36.1)	23 (36.5)	23 (41.1)
Not related	20 (32.8)	21 (33.3)	21 (37.5)
Related	5 (8.2)	3 (4.8)	11 (19.6)
Arthralgia	1 (1.6)	0(0.0)	0(0.0)
Burning sensation	1 (1.6)	0 (0.0)	0(0.0)
Contusion	0 (0.0)	0(0.0)	1(1.8)
Ecchymosis	0 (0.0)	1 (1.6)	0 (0.0)
Hypoesthesia	0 (0.0)	0(0.0)	1(1.8)
Injection-site discoloration	0 (0.0)	0(0.0)	2 (3.6)
Injection-site hemorrhage	0 (0.0)	0(0.0)	1(1.8)
Injection-site irritation	1 (1.6)	0(0.0)	2 (3.6)
Injection-site pain	1 (1.6)	1 (1.6)	2 (3.6)
Injection-site rash	1 (1.6)	0(0.0)	0 (0.0)
Joint stiffness	0 (0.0)	0(0.0)	2 (3.6)
Metatarsalgia	0 (0.0)	0(0.0)	1(1.8)
Muscle spasms	1 (1.6)	0(0.0)	1(1.8)
Pain in extremity	2 (3.3)	1 (1.6)	1(1.8)
Paresthesia	0 (0.0)	0(0.0)	1 (1.8)
Skin hyperpigmentation	0 (0.0)	0(0.0)	1 (1.8)
Intensity of TEAEs, n (%)			
Mild	20 (32.8)	20 (31.7)	19 (33.9)
Moderate	2 (3.3)	3 (4.8)	3 (5.4)
Severe	0 (0.0)	0(0.0)	1 (1.8)
TEAEs leading to discontinuation, n (%)	0 (0.0)	0 (0.0)	0 (0.0)
Most common TEAEs (>4%)*, n (%)			
Hyperkeratosis	2 (3.3)	5 (7.9)	1(1.8)
Injection-site pain	1 (1.6)	2 (3.2)	3 (5.4)
Muscle spasms	3 (4.9)	2 (3.2)	2 (3.6)
Nasopharyngitis	3 (4.9)	2 (3.2)	1(1.8)
Pain in extremity	3 (4.9)	2 (3.2)	3 (5.4)
SAE	0 (0.0)	0(0.0)	1(1.8)
AESI, n (%)	1 (1 0)	0 (0 0)	0 (0 0)
Hypersensitivity	1 (1.6)	0(0.0)	0(0.0)
Remote spread of toxin	0(0.0)	0(0.0)	0(0.0)

Abbreviations: AboBoNT-A, abobotulinumtoxinA; AESI, adverse event of special interest; n, number of participants; SAE, serious adverse event; TEAE, treatment-emergent adverse event.

Data are for the safety population and are shown for the double-blind phase only.

* Cut-off applies to all treatment groups.

presentation at baseline). Time to retreatment was also longer in the aboBoNT-A 300U or 500U groups compared with placebo, using a priori retreatment criteria based on clinical symptomatology. This suggests that aboBoNT-A may offer an advantage in the need for pain relief treatment. Furthermore, NNT scores (clinical response and MCID) were shown to be similar to those achieved with numerous therapies for chronic pain conditions (50,51), suggesting that the degree of pain reduction observed following treatment in this trial is equivalent to that observed with well-established pain medications. The MCID values were based on those reported previously (as measured by the NPRS in post-bunionectomy pain trials) (32,33,46), suggesting that, for those patients who reached the MCID, this was a clinically relevant finding. These multiple lines of evidence support the conclusion that BoNT-As may be a beneficial intervention for pain relief in HV.

Pain evaluation is subjective and many pain-rating methods are currently in use, including the Visual Analogue Scale (VAS), the Manchester Foot Pain and Disability Index, and the FFI (37,52-54). The NPRS was used in the present study because it is widely applied in both research and therapeutic settings, and is also one of the few validated unidimensional measures of self-reported pain (55), and is recognized by global health authorities as such. Although continuous rating scales are widely used in both research and clinical practice, they may not adequately describe the treatment benefit in individuals and may result in misleading conclusions about a patient's perception of pain relief. For example, studies in fibromyalgia, acute pain and arthritis often result in U-shaped distribution efficacy curves, with participants either achieving excellent pain relief or little to none, thereby describing the actual experience of almost nobody in the trial (50). More recently, alternative, dichotomous methods for evaluating pain have emerged. For example, in various pain disorders, including musculoskeletal disorders, a mean reduction of approximately 10% to 30% on the NPRS is representative of an MCID in pain intensity, thereby underscoring its clinical significance (56,57). This corresponds to approximately 1 to 3 points on the 11-point NPRS and thus served as the basis for the priori MCID in this trial (1.5 points on NPRS), which was homologous to the NPRS MCID observed in several bunionectomy trials (32,33,46), as well as a related foot pain syndrome of plantar fasciitis (25). In the current study, we detected differences between treatment groups using 2 alternative, dichotomous measures based on the NPRS to assess the number of days spent in a "reduced pain state," which supported the results of responder analyses. Dichotomous measures may be more relevant for clinicians and patients in the assessment of pain in HV than weekly mean changes on a continuous scale such as the NPRS.

With the exception of small pharmacodynamic studies, such as those involving BoNT injections to the extensor digitorum brevis muscle (58), there is a paucity of published data on the effects of BoNT-A injections into the foot for the control of foot pain or dysfunction. A significant reduction in foot pain (Pain VAS and Pain Relief VAS) at 3 and 8 weeks post-injection was reported in a small study of 27 participants with plantar fasciitis injected with BoNT-A compared with saline in the contralateral foot (25). In one of the first studies directly evaluating changes in foot function in HV following BoNT-A treatment, Radovic and Shah reported clinically significant reductions in pain as well as reductions in HV angle (27). Moreover, in a small, placebo-controlled study of participants with mild/moderate HV, foot pain (as measured by the FFI) was significantly reduced after a single injection into selected forefoot muscles from 8 weeks that persisted for up to 6 months (28); results of the present study support and extend these findings, showing that long-term pain control may be achievable in HV with repeated BoNT-A injections.

In addition to HV-related pain, patients with HV experience significant impairments in their disability status or in general mobility (8,9). Although the results from the FFI disability scale and FFI activity limitation scale showed nominal improvement over time in these domains following treatment with aboBoNT-A compared with placebo, the changes were not statistically significant and thus suggest that the impact of aboBoNT-A is limited. The present study is the first, to our knowledge, to evaluate disability outcomes in HV using the PGI-S and PGI-I scales. The failure to detect a change in disability status may be due to lack of efficacy for either aboBoNT-A dose or, alternatively, the PGI may be insensitive to subtle changes in HV disability symptoms and thus may not be an ideal tool to evaluate functional changes in this population. The failure of aboBoNT-A to demonstrate benefit in HVrelated disability is in line with the only other study to date to evaluate disability in participants with HV following BoNT-A treatment (28), but is in contrast to that reported with BoNT-A treatment in plantar fasciitis, which significantly improved overall foot function compared with placebo (25). Further work is needed to evaluate the impact of BoNT-A treatment on disability associated with HV.

Although pain remains a central consideration for HV diagnosis and treatment, an endpoint of significant clinical interest is the degree to which the lateral angular deviation of the hallux is reduced by treatment. Addressing morphological changes in HV is typically the principal goal of HV surgery, and involves returning the hallux to midline or approximating normal HV and IM angular ranges (0-15 degrees for HV and <12 degrees for IM). This is typically achieved via operative resection and structural stabilization of the hallux. Therefore, it remains important to compare the degree to which any non-operative treatment can achieve

this morphological "gold standard." In the present trial, a small reduction in HV angle (-0.99 degrees) was observed with aboBoNT-A 300U compared with placebo at week 12. Although this was not clinically significant, it may reflect a lack of worsening in the aboBoNT-A 300U group, because participants treated with placebo in the present study showed an increase in HV angle (+0.30 degrees) by the same week 12 time point. This suggests that more time may be required to show an impact of abo-BoNT-A treatment on HV and IM angle reduction. In contrast, other studies have shown substantial improvement in angular deviation over time with BoNT-A compared with placebo (26,27).

In terms of safety, the AE profile observed in this trial was largely similar to that reported by participants treated with placebo, and the safety profile was in line with the known profile of aboBoNT-A (29,30). The most frequently reported TEAEs for participants who received aboBoNT-A (300U or 500U) during the double-blind period were hyperkeratosis (5.0%), pain in extremity (4.2%) and injection-site pain (4.2%). The most frequently reported TEAEs in the placebo group were pain in extremity, muscle spasms and nasopharyngitis (4.9% each). There was no pattern indicating a unique safety signal following treatment in this study.

The study is subject to a number of potential limitations, some of which reduce the generalizability of the findings. The study population was limited to nonsurgical participants with mild to moderate angular deformity and without arthritic immobility or hypermobility. In addition, only general forefoot pain was assessed. It therefore remains to be determined whether specific types of HV pain (e.g., medial eminence pain, metatarsalgia, first metatarsophalangeal pain, midfoot pain) are more responsive to aboBoNT-A. However, because the underlying mechanisms of pain in disparate foot regions are mediated by the general nociceptive principles of localized inflammation and subsequent afferent pain signal propagation (59), it is plausible that the efficacy signal observed in this trial could be reproduced in other neuromuscular podiatric pain conditions. Another potential limitation in the generalizability of the findings in the trial is that only one foot per participant was treated, thus pain reporting may have been confounded by undiscernible and multiple pain foci in the contralateral foot in patients with bilateral HV, which represents a sizable portion of the global HV population. It is also noted that participants were not required to wear the same or a certain type of shoe over the study period, thus making the variability in shoe wear among patients during the trial a potential confounding factor, with those opting for looser fitting shoes potentially more likely to report lower pain scores during the trial. Lastly, the superiority of the aboBoNT-A 500U dose group at week 12 compared with placebo on the NPRS corresponded to a 2point advantage compared with placebo on the 11-point NPRS used as the primary endpoint scale in this trial (approximately 20% change), which, while very similar to the observed perceived pain benefit in bunionectomy trials (32,33,46), may not be clinically relevant to some patients suffering from mild to moderate HV. The study provides valuable insights in an area where large, randomized, controlled trials are lacking and further clinical studies are warranted.

In conclusion, although the primary endpoint (mean change in NPRS from baseline at week 8) was not met, several other lines of evidence from the present trial, including robust efficacy at week 12, higher efficacy signals observed at the aboBoNT-A 500U dose, the proportion of treatment responders and the number of days spent in a "reduced pain state" compared with placebo, suggest that pain associated with HV may be ameliorated by treatment with aboBoNT-A 500U in participants who have not undergone HV surgery. Further clinical evaluation is needed to establish whether BoNT-A represents a viable non-operative treatment option for HV-associated pain.

Authors' Contributions

Substantial contributions to study design and data acquisition: DGA, LAD, BB, MV and RS. Data analysis or interpretation: MV and SGP.

Drafting of the publication or revising it critically for important intellectual content: all authors. Final approval of the publication: all authors.

Data Sharing

When participant data can be anonymized, Ipsen will share all individual participant data that underlie the results reported in this article with qualified researchers who provide a valid research question. Study documents, such as the study protocol and clinical study report, are not always available. Proposals should be submitted to DataSharing@Ipsen. com and will be assessed by a scientific review board. Data are available beginning 6 months and ending 5 years after publication; after this time, only raw data may be available.

Acknowledgments

The authors thank all participants involved in the study, as well as their caregivers, care team and the investigators and research staff who comprised the DYSTANCE study team. The authors also thank Nicola Winstone, DPhil, of Ashfield MedComms, an Ashfield Health company, for providing medical writing support, which was sponsored by Ipsen (Boulogne-Billancourt, France) in accordance with Good Publication Practice guidelines.

Supplementary Materials

Supplementary material associated with this article can be found in the online version at https://doi.org/10.1053/j.jfas.2022.07.002.

References

- 1. Nix S, Smith M, Vicenzino B. Prevalence of hallux valgus in the general population: a systematic review and meta-analysis. J Foot Ankle Res 2010;3:21.
- 2. Ayub A, Yale SH, Bibbo C. Common foot disorders. Clin Med Res 2005;3:116–119.
- Fraissler L, Konrads C, Hoberg M, Rudert M, Walcher M. Treatment of hallux valgus deformity. EFORT Open Rev 2016;1:295–302.
- Arinci Incel N, Genç H, Erdem HR, Yorgancioglu ZR. Muscle imbalance in hallux valgus: an electromyographic study. Am J Phys Med Rehabil 2003;82:345–349.
- Perera AM, Mason L, Stephens MM. The pathogenesis of hallux valgus. J Bone Joint Surg Am 2011;93:1650–1661.
- Wülker N, Mittag F. The treatment of hallux valgus. Dtsch Arztebl Int 2012;109:857– 868.
- Jastifer JR, Coughlin MJ, Doty JF, Stevens FR, Hirose C, Kemp TJ. Sensory nerve dysfunction and hallux valgus correction: a prospective study. Foot Ankle Int 2014;35:757–763.
- Hurn SE, Vicenzino BT, Smith MD. Correlates of foot pain severity in adults with hallux valgus: a cross-sectional study. J Foot Ankle Res 2014;7:32.
- Menz HB, Morris ME, Lord SR. Foot and ankle characteristics associated with impaired balance and functional ability in older people. J Gerontol Series A Biolog Sci Med Sci 2005;60:1546–1552.
- Gilheany MF, Landorf KB, Robinson P. Hallux valgus and hallux rigidus: a comparison of impact on health-related quality of life in patients presenting to foot surgeons in Australia. | Foot Ankle Res 2008;1:14.
- Lazarides SP, Hildreth A, Prassanna V, Talkhani I. Association amongst angular deformities in hallux valgus and impact of the deformity in health-related quality of life. Foot Ankle Surg 2005;11:193–196.
- Menz HB, Roddy E, Thomas E, Croft PR. Impact of hallux valgus severity on general and foot-specific health-related quality of life. Arthritis Care Res 2011;63:396–404.
- Yamamoto Y, Yamaguchi S, Muramatsu Y, Terakado A, Sasho T, Akagi R, Endo J, Sato Y, Takahashi K. Quality of life in patients with untreated and symptomatic hallux valgus. Foot Ankle Int 2016;37:1171–1177.
- Schrier JC, Palmen LN, Verheyen CC, Jansen J, Koëter S. Patient-reported outcome measures in hallux valgus surgery. A review of literature. Foot Ankle Surg 2015;21:11–15.
- The American Orthopaedic Foot & Ankle Society (AOFAS). Position statement: patient-reported outcomes measures. 2018. Available at: https://www.aofas.org/ docs/default-source/research-and-policy/patient-reported-outcomes-measures-position-statement.pdf?sfvrsn=c866ee8e_2. Accessed May 17, 2022.
- Ferrari J, Higgins JP, Prior TD. Interventions for treating hallux valgus (abductovalgus) and bunions. Cochrane Database Syst Rev 2004:CD000964. https://doi.org/10.1002/ 14651858.CD000964.pub2.
- Torkki M, Malmivaara A, Seitsalo S, Hoikka V, Laippala P, Paavolainen P. Surgery vs orthosis vs watchful waiting for hallux valgus: a randomized controlled trial. JAMA 2001;285:2474–2480.

- 18. Ferrari J. Bunions. BMJ Clin Evid 2009;2009:1112.
- Beck D, Raikin S, Park A. Recurrent hallux valgus: 15 year single surgeon series. Foot Ankle Orthop 2017;2. https://doi.org/10.1177/2473011417S000116. 2473011417S2473000116.
- Mittal SO, Jabbari B. Botulinum neurotoxins and cancer a review of the literature. Toxins 2020;12:32. https://doi.org/10.3390/toxins12010032.
- Winner BM, Bodt SML, McNutt PM. Special delivery: potential mechanisms of botulinum neurotoxin uptake and trafficking within motor nerve terminals. Int J Mol Sci 2020;21:8715.
- 22. Oh HM, Chung ME. Botulinum toxin for neuropathic pain: a review of the literature. Toxins 2015;7:3127–3154.
- Sprenger T, Viana M, Tassorelli C. Current prophylactic medications for migraine and their potential mechanisms of action. Neurotherapeutics 2018;15:313–323.
- 24. Rahman SHA, Mohamed MS, Hamed AM. Efficacy and safety of Nd:YAG laser alone compared with combined Nd:YAG laser with intralesional steroid or botulinum toxin A in the treatment of hypertrophic scars. Lasers Med Sci 2020;36:837–842.
- Babcock MS, Foster L, Pasquina P, Jabbari B. Treatment of pain attributed to plantar fasciitis with botulinum toxin A: a short-term, randomized, placebo-controlled, double-blind study. Am J Phys Med Rehabil 2005;84:649–654.
- Moghtaderi A, Dehghan F, Mousavizadeh A, Khakpour N. Evaluation of the therapeutic effect of botulinum toxin A on hallux valgus deformity and pain. Int J Phys Med Rehabil 2017;5:1000438.
- Radovic PA, Shah E. Nonsurgical treatment for hallux abducto valgus with botulinum toxin A. J Am Podiatr Med Assoc 2008;98:61–65.
- Wu KP, Chen CK, Lin SC, Pei YC, Lin RH, Tsai WC, Fuk-Tan Tang S. Botulinum toxin type A injections for patients with painful hallux valgus: a double-blind, randomized controlled study. Clin Neurol Neurosurg 2015;129(Suppl 1):S58–S62.
- Ipsen Biopharm Ltd. Dysport prescribing information. 2019. Available at: https:// www.accessdata.fda.gov/drugsatfda_docs/label/2019/125274s115lbl.pdf. Accessed April 6, 2021.
- Ipsen Ltd. Dysport summary of product characteristics. 2020. Available at: https:// www.medicines.org.uk/emc/product/7261/smpc. Accessed April 6, 2021.
- Ipsen Ltd. Azzalure summary of product characteristics. 2019. Available at: https:// www.medicines.org.uk/emc/product/6584/smpc/. Accessed April 6, 2021.
- Daniels SE, Baum DR, Clark F, Golf MH, McDonnell ME, Boesing SE. Diclofenac potassium liquid-filled soft gelatin capsules for the treatment of postbunionectomy pain. Curr Med Res Opin 2010;26:2375–2384.
- 33. Golf M, Daniels SE, Onel E. A phase 3, randomized, placebo-controlled trial of DepoFoam[®] bupivacaine (extended-release bupivacaine local analgesic) in bunionectomy. Adv Ther 2011;28:776–788.
- Dysport prescribing information. 2019. Available at: https://www.accessdata.fda.gov/ drugsatfda_docs/label/2019/125274s115lbl.pdf. Accessed November 2, 2021.
- Dysport summary of product characteristics. 2020. Available at: https://www.medicines.org.uk/emc/product/7261/smpc. Accessed November 2, 2021.
- **36.** Strout TD, Burton JH. Clinically significant change in physician-assigned numeric pain rating scale scores. Am J Emerg Med 2004;22:243–245.
- Budiman-Mak E, Conrad KJ, Mazza J, Stuck RM. A review of the foot function index and the foot function index – revised. J Foot Ankle Res 2013;6:5. https://doi.org/ 10.1186/1757-1146-6-5.
- SooHoo NF, Samimi DB, Vyas RM, Botzler T. Evaluation of the validity of the Foot Function Index in measuring outcomes in patients with foot and ankle disorders. Foot Ankle Int 2006;27:38–42.
- Perrot S, Lantéri-Minet M. Patients' Global Impression of Change in the management of peripheral neuropathic pain: clinical relevance and correlations in daily practice. Eur J Pain 2019;23:1117–1128.
- 40. Diep D, Ko J, Lan J, Koprowicz KT, Ko G. Benefits, safety, and adjunct modality prevalences of long-term botulinum toxin injections for cervical dystonia and myofascial neck pain: a retrospective cohort study. J Pain Res 2020;13:1297–1304.
- Levesque A, Ploteau S, Michel F, Siproudhis L, Bautrant E, Eggermont J, Rabischong B, Volteau C, Perrouin-Verbe MA, Labat JJ. Botulinum toxin infiltrations versus local

anaesthetic infiltrations in pelvic floor myofascial pain: multicenter, randomized, double-blind study. Ann Phys Rehabil Med 2021;64:101354.

- Contopoulos-Ioannidis DG, Karvouni A, Kouri I, Ioannidis JP. Reporting and interpretation of SF-36 outcomes in randomized trials: systematic review. BMJ 2009;338: a3006.
- 43. Dawson J, Doll H, Coffey J, Jenkinson C. Responsiveness and minimally important change for the Manchester-Oxford foot questionnaire (MOXFQ) compared with AOFAS and SF-36 assessments following surgery for hallux valgus. Osteoarthritis Cartilage 2007;15:918–931.
- Laucis NC, Hays RD, Bhattacharyya T. Scoring the SF-36 in orthopaedics: a brief guide. J Bone Joint Surg Am 2015;97:1628–1634.
- 45. Coughlin MJ, Saltzman CL, Nunley JA. Angular measurements in the evaluation of hallux valgus deformities: a report of the ad hoc committee of the American Orthopaedic Foot & Ankle Society on angular measurements. Foot Ankle Int 2002;23:68–74.
- 46. Riff DS, Duckor S, Gottlieb I, Diamond E, Soulier S, Raymond G, Boesing SE. Diclofenac potassium liquid-filled soft gelatin capsules in the management of patients with postbunionectomy pain: a phase III, multicenter, randomized, double-blind, placebo-controlled study conducted over 5 days. Clin Ther 2009;31:2072–2085.
- DeMets DL, Lan KKG. Lan-DeMets Alpha-spending function. In: D'Agostino RB, Sullivan L, Massaro J, eds. Wiley Encyclopedia of Clinical Trials, Online: John Wiley & Sons, 20071–10.
- Cook RJ, Sackett DL. The number needed to treat: a clinically useful measure of treatment effect. BMJ 1995;310:452–454.
- Laupacis A, Sackett DL, Roberts RS. An assessment of clinically useful measures of the consequences of treatment. N Engl J Med 1988;318:1728–1733.
- Moore RA, Wiffen PJ, Derry S, Toelle T, Rice AS. Gabapentin for chronic neuropathic pain and fibromyalgia in adults. Cochrane Database Syst Rev 2014;2014: CD007938. https://doi.org/10.1002/14651858.CD007938.pub2.
- 51. Finnerup NB, Haroutounian S, Baron R, Dworkin RH, Gilron I, Haanpaa M, Jensen TS, Kamerman PR, McNicol E, Moore A, Raja SN, Andersen NT, Sena ES, Smith BH, Rice ASC, Attal N. Neuropathic pain clinical trials: factors associated with decreases in estimated drug efficacy. Pain 2018;159:2339–2346.
- Reinoso-Cobo A, Gijon-Nogueron G, Caliz-Caliz R, Ferrer-Gonzalez MA, Vallejo-Velazquez MT, Miguel Morales-Asencio J, Ortega-Avila AB. Foot health and quality of life in patients with rheumatoid arthritis: a cross-sectional study. BMJ Open 2020;10: e036903.
- Chang C, Wang QF, Guo JC, Li DD, Fan YB, Wen JM. The biomechanical relationship between hallux valgus deformity and metatarsal pain. J Healthc Eng 2020;2020:8929153. https://doi.org/10.1155/2020/8929153.
- van der Zwaard BC, Terwee CB, Roddy E, Terluin B, van der Horst HE, Elders PJ. Evaluation of the measurement properties of the Manchester foot pain and disability index. BMC Musculoskel Disord 2014;15:276.
- 55. Hjermstad MJ, Fayers PM, Haugen DF, Caraceni A, Hanks GW, Loge JH, Fainsinger R, Aass N, Kaasa S. Studies comparing Numerical Rating Scales, Verbal Rating Scales, and Visual Analogue Scales for assessment of pain intensity in adults: a systematic literature review. J Pain Symptom Manage 2011;41:1073–1093.
- 56. Olsen MF, Bjerre E, Hansen MD, Hilden J, Landler NE, Tendal B, Hróbjartsson A. Pain relief that matters to patients: systematic review of empirical studies assessing the minimum clinically important difference in acute pain. BMC Med 2017;15:35.
- Salaffi F, Stancati A, Silvestri CA, Ciapetti A, Grassi W. Minimal clinically important changes in chronic musculoskeletal pain intensity measured on a numerical rating scale. Eur J Pain 2004;8:283–291.
- Pons L, Vilain C, Volteau M, Picaut P. Safety and pharmacodynamics of a novel recombinant botulinum toxin E (rBoNT-E): results of a phase 1 study in healthy male subjects compared with abobotulinumtoxinA (Dysport[®]). J Neurol Sci 2019;407:116516.
- Matak I, Bölcskei K, Bach-Rojecky L, Helyes Z. Mechanisms of botulinum toxin type A action on pain. Toxins 2019;11:459.